Author’s response to reviews

Title: Effects of Greenselect Phytosome® on weight maintenance after weight loss in obese women: a randomized placebo-controlled study

Authors:

Luisa Gilardini (l.gilardini@auxologico.it)
Lucia Pasqualinotto (l.pasqualinotto@auxologico.it)
Francesco Di Pierro (f.dipierro@vellejaresearch.com)
Paolo Risso (paolo.risso@mac.com)
Cecilia Invitti (invitti@auxologico.it)

Version: 2 Date: 27 Jun 2016

Author’s response to reviews:

Thank you for giving us the opportunity to revise our paper entitled “Effects of Greenselect Phytosome® on weight maintenance after weight loss in obese women: a randomized placebo-controlled study”.

As suggested by Reviewer 1, the manuscript has been edited by a native english-speaking person.

Following the Editorial comments, we confirm that the data deposited in LabArchives are publicly available and do not contain any sensitive or identifiable patient data. In addition we removed the CONSORT checklist for our study.

We hope that you now find this new, revised version acceptable for publication.

Your Sincerely

Dr Luisa Gilardini